Author's response to reviews

Title: Auditory stimulation of Opera music induced prolongation of murine cardiac allograft survival and maintained generation of regulatory CD4+CD25+ cells

Authors:

Masateru Uchiyama (mautiya@yahoo.co.jp)
Xiangyuan Jin (jinxiangyuan@hotmail.com)
Qi Zhang (chyoki@hotmail.com)
Toshihito Hirai (toshihito1308@gmail.com)
Atsushi Amano (a-amanop@juntendo.ac.jp)
Hisashi Bashuda (bashuda@juntendo.ac.jp)
Masanori Niimi (mniimi@zaf.aff.ne.jp)

Version: 2 Date: 19 January 2012

Author's response to reviews:

January 19, 2012

Vipin Zamvar, MD
Editors-in-Chief
Royal Infirmary of Edinburgh

Dear Dr. Zamvar:

Please find the attached manuscript, “Auditory stimulation of Opera music induced prolongation of murine cardiac allograft survival and maintained generation of regulatory CD4+CD25+ cells,” which we are submitting for consideration for publication as a “Research article” in Journal of Cardiothoracic Surgery.

According to the two reviewers’ advices, we appended required experiments and added the results to the manuscript.

The manuscript has been reviewed and approved by all authors. The paper has not previously been published and is not currently under consideration by another journal. The authors report no conflict of interest.

Correspondence regarding the manuscript should be directed to Dr. Masanori Niimi, Department of Surgery, Teikyo University, 2-11-1 Kaga, Itabashi-ku, Tokyo 173-8605, Japan; telephone: ++81-3-3964-1228; fax: ++81-3-3962-2128; e-mail: mniimi@med.teikyo-u.ac.jp.

Thank you for your consideration.

Sincerely,